Author's response to reviews

Title: Community acquired mastitis due to Mycobacterium abscessus: a case report

Authors:

Maria Bruna Pasticci Dr (pasticci@unipg.it)
Luigi Maria Lapalorcia Dr (luly1978@hotmail.it)
Giacomo Antonini Dr (giacomantonini@yahoo.it)
Antonella Mencacci Dr (mencacci@unipg.it)
Rosanna Mazzolla Dott (rmazzolla@hotmail.com)
Franco Baldelli Dr (rprof2f1@unipg.it)

Version: 4 Date: 14 January 2009

Author's response to reviews:

A revised manuscript has been re-submitted, please take this version in consideration for evaluation.

The changes consists on suggestion from Rewiever but also we did find that there was an error reported in the paper consisting in the fact that while the patient received cefotaxime, cefoxitin was tested in in vitro susceptibility test. Therefore we are missing cefotaxime activity. Consequently there have been changes where required in the test.

With best regards,

M Bruna Pasticci